Author's response to reviews

Title: Withdrawal of life-support in paediatric intensive care - a study of time intervals between discussion, decision and death

Authors:

   Felix Oberender (felix.oberender@rch.org.au)
   James Tibballs (james.tibballs@rch.org.au)

Version: 2 Date: 14 January 2011

Author's response to reviews:

.